We greatly appreciate your kind comments. We have highlighted the changed parts in yellow and provided our point-by-point responses to the reviewers’ comments below.

Reviewer: #1
Comments to the Author
Thank you for a great read. This is an important case as it will provide guidance for use of tolvaptan in a rare case like this. Please see file for one suggestion of using words instead of numbers in one sentence (one for 1 and five for 5; is is customary to use words for numbers under 5).
> Thank you for your comment. In the CASE PRESENTATION paragraph, corrected 1 and 5 to one and five, respectively (P5 L23).

Reviewer: #2
Comments to the Author
1. The MRI image quality needs to be improved (the bottom images are blurring and hard to see whether normal or not), and comment briefly who this is different from normal infant.
> Thank you for your comment. Sorry, but MRI is a magnified image and therefore has limited resolution. A triangle was added to Figure 1D to show the ectopic posterior lobe of the pituitary gland (P15 L7-8). The posterior lobe with a high signal on T1-weighted images is smaller and shifted upward than normal, as noted in the text (P6 L17-19)

2. For Figure 2, if the normal or expected temporal trajectory of water intake could be shown as reference, that will help the readers understand the problem and the treatment effect.
> Thank you for your comment. A general guideline for water intake has been added to Figure 2 as a gray band (P16 L3-4).

3. Background could be enhanced with the clear purpose of the study and also address limitations of the study in the Discussion section.
> Thank you for your comment. The purpose was added to the BACKGROUND and the discussion (P3 L9-12, P8 L10-12). The limitations of the study were also added to the discussion (P9 L20-24).

4. The Imaging examination methods and Outcome need more details.
> Thank you for your comment. MRI imaging methods have been added (P6 L14-16).

5. References number (less than 15) seems not enough, especially in the Introduction section.
> Thank you for your comment. More references and details are given in the introduction (P5 L7-9).